Author's response to reviews

Title: Diffusion tensor imaging differences relate to memory deficits in diffuse traumatic brain injury

Authors:

Eva M Palacios (evapalacios@ub.edu)
Davinia Fernandez-Espejo (daviniafernandez@ub.edu)
Carme Junque (cjunque@ub.edu)
Rocio Sanchez-Carrion (rsanchezcarriion@guttmann.com)
Teresa Roig (troig@guttmann.com)
Jose M Tormos (jtormos@guttmann.com)
Nuria Bargallo (bargallo@clinic.ub.es)
Pere Vendrell (pvendrell@ub.edu)

Version: 4 Date: 20 February 2011

Author's response to reviews:

Prof. Melissa Norton
Editor-in-Chief
BMC Neurology

Dear Prof. Norton,

Please find enclosed the revised version of our manuscript. We have made all the formatting changes requested by your editorial team:

References: We made a few corrections and make sure references are cited in numerical order in the text and in the reference list.

Tables: We have re-layout the tables’ section without merged elements.

Yours sincerely,
Carme Junqué, PhD